



Xiuli Sun<sup>1,\*</sup>, Cui Zhou<sup>1</sup> and Zhuojiong Gan<sup>2</sup>



<sup>2</sup> Department of Economics, Xi'an Jiaotong-Liverpool University, Suzhou 215123, China

\* Correspondence: sunxl@swufe.edu.cn

Abstract: While the literature has examined the key role of green finance policy on firms' green innovation and environmental performance, little attention has been paid to firms' environmental, social, and governance (ESG) performance, which is increasingly important to stakeholders. Exploiting heterogeneity in firms' exposure to the green finance pilot zones policy in China in 2017 as a quasi-natural experiment, this paper employs the difference-in-differences model to explore the effect of green finance policy on firms' ESG performance. Based on the data of listed manufacturing firms in China during 2013–2020, our results indicate that the green finance policy could promote firms' ESG performance. Moreover, the overall positive effect is driven mainly by the environmental pillar. Utilizing subsample estimation and the triple differences method, we further find that the higher ESG performance is driven by firms with less financial constraints, firms in economically more developed pilot zones, and state-owned enterprises (SOEs). Mechanism analysis indicates that the pilot policy promotes firms' ESG performance even if it worsens firms' financial constraints. Our study contributes to the research on both the impacts of green finance policy and the relationship between financial constraints and ESG performance, as well as to the literature on ESG structure.

Keywords: green finance; ESG; CSR; DID; financial constraints; triple differences



Citation: Sun, X.; Zhou, C.; Gan, Z. Green Finance Policy and ESG Performance: Evidence from Chinese Manufacturing Firms. *Sustainability* 2023, *15*, 6781. https://doi.org/ 10.3390/su15086781

Academic Editor: Eftichios Sartzetakis

Received: 28 February 2023 Revised: 10 April 2023 Accepted: 15 April 2023 Published: 17 April 2023



**Copyright:** © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/).

# 1. Introduction

Technological change has greatly raised the living standards of humans. Clearly, humans have been successful in utilizing new technologies. However, we are meeting great challenges in addressing environmental damage and climate change [1]. During the period 2000–2019, there were 7348 massive disaster events resulting from extreme weather around the world, leading to 1.23 million deaths and USD 2.97 trillion in economic losses [2]. World leaders have come to a general consensus on environmental protection. The 2021 United Nations Framework Convention on Climate Change 26th Conference of the Parties (COP26) promised to increase funds for developing countries in order to tackle climate change.

As an economically fast-growing developing country, China accounts for a third of the world's greenhouse gases and 27% of global carbon dioxide [3]. Meanwhile, China is making a great effort to transition to a greener economy. To reach peak carbon emissions before 2030 and achieve carbon neutrality by 2060, China will require a massive shift in resources and new technologies to reduce pollution emissions and enhance energy efficiency and resource productivity. However, limited financial resources might largely hinder the investment of firms in green technology [4]. According to the World Economic Forum's estimation, China needs to close an annual funding gap of about RMB 1.1 trillion (USD 170 billion) [5]. Green finance is designed to make sustainability a part of every firm's financial decision making, and thus, it is promoted by governments and financial institutions, including central banks [6] worldwide in their effort to move towards sustainable development. There is also a continuously increasing interest among academics to evaluate the effectiveness of green finance in promoting green transition [4,7,8].

Because of the existence of externalities, firms are inclined to make excessive investments in polluting industries and inadequate investments in green projects. Economically, the aim of green finance policy is to use policy and institutional arrangements to address the positive externalities of green investments or negative externalities of polluting investments, which cannot be internalized solely by the market [9].

As the green finance system develops and its institutional structure improves, firms are under pressure to disclose environmental, social, and governance (ESG) performance. Firms are being evaluated by sustainability rating agencies such as Thomson Reuters, MSCI, and Bloomberg. ESG ratings are the result of an assessment of a firm's quality, standard, or performance on ESG issues [10]. Sustainable and responsible investors depend strongly on the ESG scores provided by these rating agencies [11]. ESG performance now serves as a critical measurement of corporate sustainability, especially among investors, corporations, and policymakers [12]. Consistent with the increasing importance of ESG performance, more academic studies have tried to identify underlying mechanisms that explain why firms differ in their ESG performance [11,13–16].

ESG means environmentally and socially friendly activities not merely required by law but which go beyond compliance as well, privately supplying public goods, or voluntarily internalizing externalities. Firms adopt different ESG strategies to maximize firm value and minimize costs and risks in the long run. Among all the factors influencing firms' corporate social responsibility (CSR) engagement, Campbell proposed that firms in weak financial situations are less likely to invest in social responsibility [17]. Nevertheless, empirically identifying a causal link running from financial resources to firms' stewardship toward ESG is particularly difficult since unobservable factors may jointly determine a firm's financial resources and its ESG engagement [18].

This paper examines whether green finance affects firms' ESG engagement. To identify, we exploit heterogeneity in firms' exposure to green finance policy in China and pin down the causal impact of green finance on firms' ESG engagement. Green finance seeks to provide financing, operating funds, investments, and other financial services for eco-friendly projects, with ecological preservation as the key driver [19,20]. Whether green finance plays a role in the ESG performance of the corporate sector is still an open question.

Existing literature reveals the close association between green finance activities and sustainable development at both macro and micro levels [20–23]. At the macroeconomic level, for example, based on the data of 25 provinces in China from 2004–2017, Zhang and Wang constructed an evaluation system to assess green finance growth using the Pressure-State-Response model and revealed that green finance could foster sustainable energy development [22]. On the other hand, Liu and Wang provided micro evidence of green finance impacts [20]. Using Chinese listed companies between 2013 and 2020, they demonstrated a robust link between green finance and growing green patent applications.

Green finance studies at the macro level are usually based on the green finance development index, e.g., Zhang and Wang [22]. However, they could easily suffer from endogeneity issues [24]. Policy experiments provide researchers with a good opportunity to explore causality. By examining the impact of green finance policy, scholars have suggested that green finance policy is related to green innovation [8,20,25], corporate investment efficiency [24], controlling the overall situation of air pollution [7], enterprise energy consumption intensity [23], debt-financing cost of heavy-polluting firms [26], and environmental pollution reduction at the macro level [27]. Earlier research focused mainly on the environmental dimension. However, up to now, few studies have explored the combined economic, social, and environmental effects of green finance at the firm level. To the best of our knowledge, the only attempt was made by Li et al., who investigated the effects of green finance policy on firms' ESG performance by examining the 2012 Green Credit Guidelines (GCG) policy in China [28]. They found that the GCG policy enhances ESG performance in firms restricted by the policy, compared to firms without the restrictions.

Our study differs from their analysis in several ways. Firstly, different green finance policies are examined. The GCG policy is designed to force the banking sectors to include environmental requirements in issuing loans to heavy-polluting firms. In other words, the only green financial instrument of the GCG policy is a green loan. On the other hand, the green finance pilot zones policy examined in our paper combines various instruments of green finance such as green bonds, green loans, green insurance, and green fintech. Secondly, Li et al. relied on a self-constructed ESG index while we use Bloomberg ESG as well as MSCI ESG to measure firms' ESG performances [28]. In social responsibility literature, conclusions are sensitive to ESG measurement [29]. The widely used Bloomberg and MSCI ESG indicators allow us to make comparisons with other ESG studies. Thirdly, since the pilot zones policy was implemented in only eight zones, the policy provides us with a natural treatment group and a control group, and thus we do not need to define them ourselves, as was performed by Li et al. [28]. Fourthly, compared with Li et al. [28], we pay particular attention to state-owned enterprises (SOEs). In China, SOEs could typically benefit from soft budget constraints. In fact, Yu et al. showed that green finance policies have primarily eased the financing constraints of SOEs, and green credits are possibly less accessible to privately owned firms (POEs) [4]. Thus, the analysis of SOEs could deepen our understanding of the effects of green finance policy.

We focus on the manufacturing firms in China since the sector has played a key role in China's fast economic expansion, and its value added contributes to about 27.44% of GDP in 2021, according to the World Bank. China serves as a perfect setting for our study. As the world's largest manufacturing power, the manufacturing industry has significant environmental impacts in China. According to China Energy Statistical Yearbook 2021, the manufacturing industry accounts for 57.9% of China's total energy consumption and produces more than 50% of total  $CO_2$  emissions. In 2019, the Chinese manufacturing industry contributed 12.24% of the world's carbon dioxide releases and 13.46% of energy consumption [30]. As the central pillar of the Chinese economy, the technological upgrading of the manufacturing sector is a vital strategic task for economic development [31].

By examining listed firms in China in 2013–2020, we study how the green finance pilot zones policy affects firms' ESG performance using the difference-in-differences (DID) estimation. First, the main result is that the pilot zones policy significantly increases the ESG performances of firms. Specifically, for firms exposed to the pilot policy, the ESG score will increase by 7.3% when evaluating the sample mean. Secondly, the overall positive effect of the green finance policy on ESG performance is driven mainly by the environmental pillar rather than the social and governance pillars. Thirdly, utilizing both subsample estimation and triple differences, we further find that the higher ESG performance is mainly driven by heavy-polluting firms, firms with less financial constraints, firms in economically more developed pilot zones, and state-owned enterprises (SOEs). Fourthly, mechanism analysis indicates that the pilot policy promotes the ESG performances of firms even if it worsens firms' overall financial constraints. The reason for this might be that the policy aims at environment-friendly projects. Finally, the results are robust to the parallel trend test, PSM-DID, alternative ESG proxy, and placebo test.

Our paper sheds light on the role of green finance in firms' sustainable performance. Scholars have noticed the key role of green finance in promoting the diffusion of environmental innovation [32], carbon mitigation [19], environmental responsibility [33], and green performance and innovation [34]. Much of the green finance literature has focused on environmental performance, but little attention has been paid to ESG performance, which is increasingly important to stakeholders [10,11]. Utilizing the Bloomberg ESG database and the data of listed manufacturing firms in China from 2013 to 2020, our results indicate that the green finance policy could promote the ESG performances of firms.

Furthermore, our findings contribute to research examining the role of financial constraints for CSR. Leong and Yang confirmed the negative effects of financial constraints on all dimensions of CSR performance [35]. Hong et al. showed that financial constraints are a critical obstacle to corporate social responsibility [18]. They showed that when firms' constraints are exogenously relaxed, firms with higher financial constraints improve their CSR performance relative to less-constrained firms. Xu and Kim found that the relaxation of financial constraints reduces US-listed firms' toxic emissions [36]. We find that the pilot policy tightens firms' financial constraints but promotes their ESG performance. The reason for this might be that the policy is targeted to improve firms' environmental performance. More importantly, the overall positive effect of the green finance policy on ESG performance is driven mainly by the environmental pillar rather than the social and governance pillars.

The paper makes an additional contribution by shedding light on the role of green finance policy on sustainable development. Previous literature has related the same pilot zones policy to environmental pollution control [27], high-quality green innovation [37], the decrease of debt-financing cost [26], the increase of green patent output [20], the reduction of inefficient and excessive investments [24], environmental quality improvement [2], overall air pollution control [7], and the reduction of energy consumption intensity [23]. Lu et al. investigated the Green Credit Guidelines in China in 2012 and found that the green finance policy increases the financial constraints and debt financing cost of high-polluting enterprises. Li et al. examined the same Green Credit Guidelines (GCG) policy. Their results showed that the green finance policy promotes restricted firms' social responsibility even though it tightens financial constraints. Restricted firms are firms restricted by GCG, and they are heavily polluting firms. We reach a similar conclusion to Li et al. by examining the green finance pilot policy in 2017. We further point out that the higher ESG performance is mainly driven by heavy-polluting firms, firms with less financial constraints, firms located in economically more developed pilot zones, and SOEs.

The remainder of this paper is organized as follows. Section 2 presents the institutional background and provides the literature review. Section 3 provides an empirical strategy. Section 4 describes the data source and descriptive statistics. Section 5 provides the empirical results, including parallel trend tests, heterogeneity analysis, and mechanism analysis. Section 6 describes the robustness tests. Section 7 provides a conclusion.

#### 2. Related Literature

## 2.1. Institutional Background

After decades of fast economic growth, China has now prioritized developing an environmentally friendly economy. In September 2020, President Xi promised that China would reach the maxima of carbon emissions by 2030 and net-zero of all six types of greenhouse gas releases and carbon neutrality by 2060 [16]. Achieving these goals involves substantial economic and social transformation and immense investment. This will necessitate a strong financial system to stimulate, facilitate, and steer finance into these areas [38].

Green finance is specified by the People's Bank of China (PBOC) as "financial services that support economic activities aimed at improving the environment, mitigating climate change, and utilizing resources more efficiently" [39]. In other words, any structured financial activity, a product or service, which has been created to stimulate the development of green projects, minimize the environmental damage of regular projects, or a combination of both, could be viewed as green finance.

China formulated the green finance development plan as early as the 1990s and has strived to develop the green finance system over the years. Green finance offers an economic means to bridge the gap between the supply and demand of sustainable economic development [7]. Though still at an early stage, China has made impressive progress in improving the green financial system and markets. China's green bond market has emerged as the world's second-largest, after the US [40]. According to PBOC, China issued 1643 green bonds until the end of the year 2021, with a total balance of RMN 1727 billion (USD 270 billion) [41].

The green finance pilot zones are part of the central government's efforts to establish a strong green financial system. The first batch was initiated in June 2017 in eight regions across eastern, central, and western parts of China, at different development stages [24]. The aim of establishing the pilot zones was to gain experience to think through scale issues

as policies are elevated nationally. Specifically, as indicated in Figure 1, they are Huzhou city and Quzhou city in Zhejiang province, Ganjiang New District (located in Nanchang city) in Jiangxi province, Guangzhou city in Guangdong province, Gui'an New District (located in Guiyang and Anshun) in Guizhou, and Hami city, Changji prefecture, and Kramay city in Xinjiang.



**Figure 1.** Eight green finance pilot zones in five provinces in China. In June 2017, eight regions in Zhejiang, Guangdong, Jiangxi, Guizhou, and Xinjiang, were selected as the first batch of pilot zones for green finance reform.

The primary goals for the pilot zones are to facilitate the establishment of green finance departments or branches within financial institutions, promote green credits, establish markets for trading rights over emission, energy use, and water, develop public service channels for green industries, and build risk management mechanisms for green finance [7]. However, the focuses of the eight regions in the five provinces differed [26], as shown in Table 1, since they were at different stages of development. For example, the Guangzhou zone aimed to develop financial products and services innovation, green fintech, and green finance markets in cooperation with Hongkong and Macao capital markets. At the same time, the Xinjiang zone focused mainly on developing green finance for agriculture modernization, clean energy, and high-end manufacturing related to renewable energy.

The pilot zones have achieved significant milestones. As of 2020, the outstanding green loans in the pilot zones reached RMB 236.8bn (USD 35bn), accounting for 15.1% of the overall lending portfolio [42]. The Green Credit Platform developed in Huzhou has assisted over 17,000 enterprises to obtain more than ¥ 187 billion in financing as of July 2020 [43]. Guangzhou has a robust carbon pilot trading market, the first green project financing integration system, and real momentum in coordinating efforts to stand up green finance in the Greater Bay Area. In fact, according to Lv et al., the green finance development of these five provinces was ranked top six, with Beijing ranking first [44].

Province	Municipality		Priorities
Zhejiang	Huzhou city Quzhou city	1. 2.	Innovating green finance to foster the transformation of traditional industries, including green credit systems, green bank rating, green insurance, etc. Green financing for micro, small, and medium enterprises
Guangdong	Guangzhou city	1. 2. 3.	Green financial products and services innovation, including establishing environmental equity trading markets, green bonds, green insurance, etc. Developing green fintech Developing green finance markets in cooperation with Hongkong and Macao capital market
Jiangxi	Ganjiang New District (located in Nanchang city)	1. 2.	Establishing fully operational green finance system with numerous products and services Upgrading traditional industries
Guizhou	Gui'an New District (located in Guiyang and Anshun cities)	1. 2. 3.	Optimizing local ecological resources in developing green finance to boost clean and sustainable economic growth Alleviating poverty Building sharing infrastructure for big data
Xinjiang	Hami City, Changji Prefecture, and Karamay City	1. 2.	Demonstrating role in the Green Belt and Road Developing green finance to foster agriculture modernization, clean energy, and high-end manufacturing related to renewable energy

#### Table 1. Priorities in pilot zones.

An additional pilot zone located in Lanzhou New District in Gansu province was announced in 2019. The priority of the Lanzhou zone is to develop a local green finance system and to promote electric vehicles and big data industries. We do not include it in this paper since we only focus on the first batch of pilots.

# 2.2. Related Literature

## 2.2.1. Factors Influencing Firm ESG Engagement

Existing studies have proposed various theories on why firms invest in ESG. Among them, the shareholder expense view and the stakeholder view are the two main views. The stakeholder view argues that CSR activities could increase shareholder wealth since the satisfaction of other stakeholders' interests makes them more supportive of the firm, which contributes to maximizing shareholder wealth. That is, stakeholder theory argues that corporate managers have to meet the need of stakeholders to maximize firm value [45,46]. Contrary to the stakeholder view, the shareholder view argues that managers' engagement in social responsibility benefits other stakeholders at the cost of shareholders. In other words, the shareholder view argues that CSR damages shareholder value [47–49].

To legitimize ESG on sound economic grounds, numerous studies have investigated the effects of ESG on firms. Prior literature has documented that ESG is positively related to firm financial performance [50–53]. Prior studies also related ESG to different aspects of firms, such as firm value [54], risk mitigation [55], insurance mechanism [56], and access to finance [57].

ESG is a strategic decision of the firm. Scholars have investigated the factors that influence firms' ESG engagement. Firm characteristics, such as ownership structure [58] and firm size [16,59–61], have been proven to be vital in firms' ESG performances in numerous studies. Academics have also started to focus on the role of corporate governance in ESG, including board structure [60,62], CEO duality [60,63], female board members [64,65], and the presence of independent directors [16,60,61,66]. The underlying philosophy is that since managers could significantly affect a firm's major decisions, they should also have a critical influence on its ESG engagement decisions. We briefly summarize the related literature on ESG performance in Table 2.

Paper	Theory	ESG Indicators	Main Findings
Baldini et al. (2018) [14]	Stakeholder theory	Bloomberg ESG; firms from Japan, the US, the UK, and Australia during the 2005–2012 period	A country's cultural system, labor system, and political system could largely affect firms' ESG disclosure.
Chan et al. (2017) [13]	Shareholder theory	MSCI ESG STAT ratings; firms listed in the US market from 1992 to 2010	The results suggested that financial constraints are negatively linked to CSR activities.
DasGupta (2022) [15]	Stakeholder theory	Thomson Reuter's Asset4 ESG ratings; firms from 27 countries during 2010–2019	The paper found that financial performance shortfall could positively affect a firm's ESG performance.
Drempetic et al. (2020) [11]	Stakeholder theory	Thomson Reuters ASSET4 ESG ratings; firms from the USA, Japan, Great Britain, and Canada during the 2004–2015 period	The ESG scores are biased towards bigger firms due to the necessity of resources for obtaining ESG data.
Galbreath (2016) [66]	Shareholder theory and stakeholder theory	Sustainable Investment Research Institute (SIRIS) CSR ratings; firms in the Australian Securities Exchange 300 (ASX300) from 2004–2009	External directors and female board members complement each other, as their interaction effect gradually affects CSR beyond their individual, independent effects
Hansen et al. (2018) [49]	Shareholder theory and stakeholder theory	MSCI ESG KLD STATS; US firms in the manufacturing sector for 1992–2009	The results show that CSR investment and its economic influences are context specific.
Liao et al. (2018) [62]	Shareholder theory and stakeholder theory	Rankins CSR ratings; listed firms over the period of 2008–2012 in China	Larger board sizes, separation of CEO and chairman positions, and more female directors are positively linked to CSR engagement.
Oh et al. (2011) [58]	Shareholder theory	KEJI index; listed Korean firms in 2005	Institutional and foreign investors tend to engage CSR more.
Shu and Tan (2023) [16]	Stakeholder theory	Hexun ESG; listed firms in China during the 2010–2019 period	Carbon control policy risk reduces firms' ESG engagement, with bank loan costs and financial constraints as underlying mechanisms
Yuan et al. (2019) [63]	Shareholder theory and stakeholder theory	MSCI STATS CSR ratings; US firms from 2003 to 2012	CEO ability enhances firms' CSR ratings.

#### Table 2. Literature summary.

However, only a few studies of ESG noticed the effect of financial resources on a firm [18,67]. Based on KLD CSR ratings and the dataset of S&P 500 firms from 1991 to 2008, Hong et al. pointed out that neglecting the heterogeneity of firms' financial constraints could bring in a spurious link between financial performance and corporate social responsibility, even if the motives for social responsibility are not-for-profit [18]. In their main identification strategy, they used a difference-indifferences method by exploiting the exogenous shock of the Internet bubble of the late 1990s. Based on MSCI ESG STAT ratings and the data of listed firms in the US market from 1992 to 2010, Chan et al. investigated the association between financial constraints and CSR, and they confirmed a significant negative relationship between financial constraints and CSR activities [13]. However, they fail to establish a causal relationship that less financially constrained firms engage more in CSR. Exploiting exogenous variation in financial constraints, Hong et al. showed that financial constraints are a critical driver of CSR [18]. In spite of the motives for CSR engagement, less constrained firms have a larger expenditure on social responsibility.

ESG spans three dimensions of a firm's activities and captures a firm's efforts to address various externalities that it produces while pursuing profit maximization that is not internalized by shareholders [68]. Thus, beyond viewing ESG as a whole, we consider different dimensions of ESG as fundamental tradeoffs among different stakeholders. The engagement activities across different ESG dimensions represent various aspects of stakeholder concerns [68]. According to the stakeholder theory, allocating firm resources

to social responsibility not directly relevant to primary stakeholders may not result in sustainable competitive advantage [69,70]. The relationship between CSR engagement and firm financial performance varies with CSR dimensions. For example, using the MSCI ESG database and annual data from COMPUSTAT from 1991 to 2012, Tsai and Wu showed that the practices of social responsibility pertaining to human rights and the environment matter primarily during a crisis period when the financial resource is finite [71]. On the other hand, firms in a financial crisis may reduce their engagement in employee relations to enhance their financial performance. Benlemlih and Bitar differentiated between primary and secondary stakeholders [72]. Based on the data of US firms during 1998–2012, their findings indicated that CSR components directly concerning firms' primary stakeholders are more critical in improving investment efficiency than those linked to secondary stakeholders that firms specialized in CSR dimensions, with about three-quarters of them centering on a single aspect [73]. The degree of specification varies with industries, and the focused dimension also differs even within industries with similar specification levels.

#### 2.2.2. Green Finance and ESG

Long-term and stable capital investment is a prerequisite for firms to conduct green production [4]. Thus, a firm's green development is limited by financial resources. A lack of financial resources could largely hinder investment in green technology for two reasons [4,74]. Moreover, compared to regular innovation, green technology transition is characterized by "double externalities" as it generates knowledge spillovers and environmental spillovers at the same time [25,75]. To alleviate financial constraints, public policies are needed to promote green economy transformation [76]. Among different public policies, green finance is a key financing tool to address environmental damage.

As to the mechanisms underlying the link between green finance and firms' environmental performance, scholars mainly focus on financing costs or financial constraints. Liu et al. established a financial computable general equilibrium model (CGE) [77]. The simulation results showed that with the implementation of the green credit policy, the average financing costs of the energy-intensive industries increase by 1.1% in the short run, decrease by 0.008% in the medium run, and increase by 0.003% in the long run [77]. By investigating the panel data of 52 green firms and 81 "two-high" (high-pollution and high-emissions) firms in China from 2001–2007, Xu and Li concluded that green finance has asymmetric impacts on heavy-polluting firms and relatively clean firms. Specifically, the green credit policy raises the debt financing cost of "two-high" firms but lowers that of green firms [78].

# 3. Empirical Models and Estimation Strategies

#### 3.1. Baseline Regressions

To study the effect of the pilot zones policy on firms' ESG performance, we utilize a difference-in-difference (DID) strategy. We take advantage of the fact that the green finance pilot zones policy is an exogenously applied policy for all firms since it is a top-down policy. Huang et al. argued that pilot zones are decided by the central government rather than the local government, which can mitigate the endogeneity concern arising from the self-selection bias [37]. Thus, the policy can be considered a quasi-natural experiment [26]. We then specify our empirical specification as:

$$ESG_{it} = \beta_0 + \beta_1 treat_i + \beta_2 post_t + \beta_3 treat_i \times post_t + \theta X_{it} + \lambda_i + \mu_t + \varepsilon_{it},$$
(1)

where  $ESG_{it}$  represents the firm *i*'s ESG performance in year *t*. *treat*<sub>i</sub> is a dummy variable that is equal to 1 if the firm is located in the pilot city and is otherwise 0. *post*<sub>t</sub> indicates whether or not the policy was implemented, and equals 1 for 2017–2020 and 0 otherwise. *treat*<sub>i</sub> × *post*<sub>t</sub> is the interaction of *treat*<sub>i</sub> and *post*<sub>t</sub> variables, and the coefficient  $\beta_3$  is the main interest of the study, which captures the causal effect of green finance policy on firms' ESG performance. Moreover,  $X_{it}$  is a vector of all the control variables, including

firm age, firm size, ROA, duality, female directors, independent directors, the share of the largest shareholder, SOE, leverage, growth, free cash flow, and financial constraint indicator. Firm size is measured as the log form of total assets. To capture the effects of macroeconomic factors, we also include the log form of city-level GDP per capita in the regression. Individual firm fixed effects, province fixed effects, industry fixed effects, and year fixed effects are also controlled in our specification.

Cash generally represents a substantial share of a firm's financial resources, allowing firms to make investments quickly. Thus, cash holding facilitates firms to avoid asymmetric information costs and transaction costs related to external financing.

The treatment and control groups need to follow parallel trends before the policy implementation to ensure the validity of the DID method. To test the validity of the parallel trends assumption, following the literature [79,80], we estimate the annual policy effects, which also capture the dynamics of the policy. We then estimate the following model:

$$ESG_{it} = \beta_0 + \sum_{t=2014}^{2020} \gamma_t (treat_i \times Year_t) + \theta X_{it} + \lambda_i + \mu_t + \varepsilon_{it},$$
(2)

where year dummies,  $Year_t$ , are interacted with the treatment indicator,  $treat_i$ , to generate a DID estimate for each year, using the year 2013 as a benchmark. We expect the postpolicy yearly effects to be significant and positive and the pre-policy yearly effects to be insignificant.

# 3.2. Triple Differences

Environmental damage is mainly caused by heavy-polluting firms. More specifically, the heavy-polluting sector is the main emitter of pollutants like nitrogen oxides (NOx), sulfur dioxide (SO<sub>2</sub>), and particulate matter (PM), which contribute about 99%, 98%, and 97% [81]. Therefore, investigation of the effects of green policy on heavy-polluting firms is particularly vital. As the prime targets of environmental regulation, the policy effects might be even more significant among heavy-polluting firms relative to non-heavy-polluting firms.

To directly test if heavy-polluting firms have a larger effect, we implement a triple differences test by separating both the treatment and control sample into heavy-polluting and non-heavy-polluting firms, defined based on heavy-polluting and non-heavy industries as in Shi et al. [26]. Our *pollution* indicator will not vary through the sample period.

Following previous studies [79,82,83], we explain the outcome variable  $ESG_{it}$  with a treatment dummy (*treat<sub>i</sub>*), a post-event dummy variable (*post<sub>t</sub>*), a heavy-polluting dummy (*pollution<sub>j</sub>*), the whole set of double interaction terms (*treat<sub>i</sub>* × *pollution<sub>j</sub>*, *pollution<sub>j</sub>* × *post<sub>t</sub>*), and *treat<sub>i</sub>* × *post<sub>t</sub>*), and the triple interaction term (*treat<sub>i</sub>* × *pollution<sub>j</sub>* × *post<sub>t</sub>*), controlling for a set of fixed effects:

$$ESG_{it} = \alpha \times treat_i \qquad \times pollution_j \times post_t + \beta_1 \times treat_i \times pollution_j + \beta_2 \times pollution_j \times post_t + \beta_3 \times treat_i \times post_t + \beta_4 X_{it} + \gamma_i + \varepsilon_{it}.$$
(3)

The specification includes all three double interaction terms to operationalize the triple differences estimator. Following the previous literature, we also include firm fixed effects ( $\gamma_i$ ) to capture firm-level time-invariant heterogeneity [84]. The critical coefficient of interest is the one on the triple interaction term,  $\alpha$ . It estimates whether the difference between the differential response of the heavy-polluting treated firms relative to their heavy-polluting control group and the differential response of the non-heavy-polluting treated firms relative to their non-heavy-polluting control group is significant after the implementation of the green finance policy.

# 4. Data

Our data on ESG were obtained from the Bloomberg database. The scores, ranging from 0.1 to 100, measure the transparency or disclosure quality, for a broad range of ESG dimensions, such as pollution waste disposal, greenhouse gas emissions, renewable energy, community relations, diversity, human rights, political donations, executive compensation, board size, independent directors, and employee turnover. The Bloomberg ESG scores summarized these aspects in three dimensions, environmental, social, and governance pillars, each with a 33% weighting [60]. Moreover, Bloomberg provides scores for each pillar, i.e., environmental, social, and governance scores.

One of the most broadly used ESG scores by institutional investors is provided by MSCI, formally known as KLD Research and Analytics. The MSCI score comprehensively evaluates each firm's ESG profile [71]. The Bloomberg ESG indicator is also extensively used in ESG studies in China [61,85–87] and in other countries [10]. Bloomberg data are more consistent than MSCI, and thus we chose Bloomberg ESG as our data source [86].

Compared to the MSCI data, the Bloomberg ESG is adjusted to different industries. Therefore, a firm is assessed using information relevant to its industry [14]. We use both the composite ESG score and the three component scores in our analysis.

We obtain firm characteristics and financial performance data from the China Stock Market and Accounting Research Database (CSMAR) to supplement our analysis. Our dataset consists of all manufacturing firms listed on the stock exchanges in Shanghai and Shenzhen between 2013 and 2020. We matched the ESG data and firm characteristics and financial performance data from CSMAR, excluding firms with special treatment (ST) type and missing data.

Table 3 summarizes descriptive statistics for the sample. The average ESG score is 21.39. The average environmental, social, and governance scores are 11.85, 23.59, and 44.43, respectively. Our ESG scores are consistent with other studies [61]. The average firm age is 18.59, ranging from four to 39. Besides, the mean value of SOE is 0.45, indicating that 45% of the sample are state-owned enterprises. The average firm size is 22.95. The average firms are moderately levered with a leverage ratio of 45%, a mean return on assets (ROA) of 0.04, and a mean share of the largest shareholder of 35.74%. The average growth rate, measured as the sales revenue growth rate, is 22%. The average number of female directors and the average number of independent directors are 2.83 and 3.27, respectively. The average CEO duality is 0.23, indicating that 23% of CEOs are the board's chairman at the same time. An average firm has a mean free cash flow level of 0.09. Furthermore, the statistics of the variables demonstrate that substantial variance exists among samples.

We use the Size-Age (SA) index proposed by Hadlock and Pierce to measure financial constraints [88]. The SA index is calculated using firm size and age, where firm size is measured by the natural logarithm of total assets and firm age by the total number of years since a firm was established. The SA index is computed as follows:

$$SA = -0.737 \times Size_{it} + 0.043 \times Size_{it}^2 - 0.040 \times Age_{it}.$$
 (4)

The average of the SA index is -3.83, ranging from -4.69 to -2.76. The correlation between the SA index and firm size in our sample is 0.2193. Notice that the SA index is convex in firm size, and it increases with firm size when the firm size is larger than 8.57. In our sample, the average firm size is 22.95, ranging from 19.55 to 27.55. Consistent with the previous studies, a larger SA index (i.e., smaller absolute value) indicates a less severe financial constraint [89–91].

As indicated by Wang et al., the development of ESG ratings in China is still in the early stage, and only about 30% of listed firms have ESG ratings on average [85]. Table 4 presents the distribution of observations across the industries. Following Shi et al. [26], we define firms in the following industries as heavy-polluting firms as indicated in Table 4. The heavy-polluting industries account for 50.3% of all the observations. Firms in other industries are classified as non-heavy-polluting firms. Table 4 also shows the average ESG

scores across different industries. Waste resources and material recovery and processing (C42) has the highest ESG score, 33.28, while another manufacturing industry (C41) has the lowest ESG. Among all the heavy-polluting industries, ferrous metal smelting and rolling pressing (C31) has the highest ESG (25.88), while rubber and plastic products (C29) has the lowest ESG (19.03). Moreover, the ferrous metal smelting and rolling pressing industry scores the highest in both the environmental dimension (E) and the social dimension (S), with scores of 16.85 and 26.95, respectively. Regarding the governance pillar, the automobile manufacturing industry (C36) scores the highest, with a score of 46.61.

Table 3.	Summary	statistics.
----------	---------	-------------

Variables	Obs	Source	Mean	Std. dev.	Min	Max
ESG	4486	WIND	21.39	7.00	1.24	64.11
Environmental dimension	4047	CSMAR	11.85	8.54	0.78	65.63
Social dimension	4473	CSMAR	23.59	9.26	3.51	77.19
Governance dimension	4486	CSMAR	44.43	5.19	3.57	64.54
Log(asset)	4486	CSMAR	22.95	1.21	19.55	27.55
Age	4486	CSMAR	18.59	5.31	4	39
RŎA	4486	CSMAR	0.04	0.12	-3.91	0.37
Duality	4486	CSMAR	0.23	0.42	0	1
Number of female directors	4486	CSMAR	2.83	1.72	0	10
Number of independent directors	4486	CSMAR	3.27	0.65	2	8
Largest shareholder proportion	4486	CSMAR	35.74	14.99	3.39	89.99
SOE	4486	CSMAR	0.45	0.50	0	1
Leverage	4486	CSMAR	0.45	0.23	0.0080	8.01
Firm growth	4486	CSMAR	0.22	1.82	-0.98	58.84
Free cash flow	4486	CSMAR	0.09	1.70	-24.44	39.69
SA	4486	CSMAR	-3.83	0.24	-4.69	-2.76
Log (city-level GDP per capita)	4194	China City Statistical Yearbook	11.38	0.49	9.54	12.28
Province-level GDP per capita growth	4486	China City Statistical Yearbook	0.07	0.03	-0.039	0.21

The units of total asset and city-level GDP per capita are RMB Yuan.

	Table 4. Number of firms and ESG over ind	ustries (Heavy-polluting industries are in bold
--	---	---

Industry Code	Industry Name	Number of Firms	ESG	Ε	S	G
C13	Agro-food processing	110	19.80	9.54	24.04	42.47
C14	Food Manufacturing	78	20.45	10.76	24.06	45.19
C15	Beverage manufacturing	193	19.83	11.34	20.04	44.48
C17	Textile industry	80	20.10	9.92	23.13	42.01
C18	Textile and garment industry	91	16.92	6.63	20.11	43.78
C19	Leather, fur, feather products, and shoe manufacturing	16	20.04	9.69	20.18	43.75
C21	Furniture manufacturing	12	18.35	7.83	21.64	42.26
C22	Paper making and paper products	78	22.07	12.46	22.33	46.52
C23	Printing and record medium reproduction	38	19.19	7.88	22.67	43.33
C24	Education and sports goods	16	20.66	8.87	27.08	41.29
C25	Petroleum processing, coking, and nuclear fuel processing industry	63	20.46	10.39	21.39	46.34
C26	Chemical raw materials and chemical products	460	21.15	11.22	23.67	44.15
C27	Medical and pharmaceutical products	511	22.52	13.26	25.68	44.02
C28	Chemical fiber industry	65	23.62	13.42	24.29	46.43
C29	Rubber and plastic products	79	19.03	7.83	22.83	42.81
C30	Nonmetallic mineral products	185	23.18	13.99	24.50	45.06
C31	Ferrous metal smelting and rolling pressing	163	25.88	16.85	26.95	45.92
C32	Nonferrous metal smelting and rolling processing	255	23.64	13.98	25.64	45.09
C33	Metallic mineral products	90	20.19	11.09	21.33	43.00
C34	General equipment manufacturing	214	20.52	10.27	24.15	43.41
C35	Special equipment manufacturing	305	20.69	10.65	23.15	44.88
C36	Automobile manufacturing	183	24.40	14.72	26.40	46.61
C37	Transportation equipment	145	20.76	11.36	20.53	45.96
C38	Electrical machinery and equipment manufacturing	380	20.57	10.83	22.69	44.01
C39	Communications equipment, computers, and other electronic equipment	583	20.80	11.70	22.83	44.39
C40	Instrument and meter manufacturing	51	18.94	8.51	21.67	43.03
C41	Other manufacturing	29	15.19	8.63	14.34	41.13
C42	Waste resources and material recovery and processing	13	33.28	26.42	39.00	43.27

# 5. Results

# 5.1. Baseline Regression Results

We empirically examine the effects of green finance by examining the green finance pilot zones policy through the DID model. Green finance might be endogeneous since policymakers might make green finance development decisions based on firm performance [39]. Compared to the ordinary least squares (OLS) method, DID can solve endogeneity problems [92]. Similarly, Li et al. investigated the effects of green finance on CSR by examining the 2012 GCG policy through the DID model [28].

Table 5 reports the effect of the green finance pilot zones policy on the ESG performance of a firm. Robust standard errors are used across all specifications. Column (1) examines the basic impact of the green finance policy on the ESG performance of firms.

	(1)	(2)	(3)	(4)	(5)
Treat * post	1.538 ***	1.600 ***	1.607 ***	1.595 ***	1.574 ***
	(0.404)	(0.404)	(0.404)	(0.408)	(0.407)
Post	0.537 ***	0.522 ***	0.491 ***	0.488 ***	2.363 *
	(0.172)	(0.180)	(0.180)	(0.180)	(1.436)
Firm size	1.082 ***	1.082 ***	1.151 ***	1.230 ***	1.217 ***
	(0.192)	(0.200)	(0.201)	(0.212)	(0.213)
Firm age	1.270 ***	1.249 ***	1.229 ***	1.249 ***	0.971 ***
C C	(0.0911)	(0.0945)	(0.0939)	(0.0956)	(0.231)
ROA	0.261	0.462	0.485	0.506	0.565
	(0.548)	(0.575)	(0.564)	(0.563)	(0.566)
Duality	0.113	0.0934	0.0893	0.145	0.135
-	(0.207)	(0.209)	(0.209)	(0.211)	(0.211)
Female directors	0.0711	0.126 **	0.131 **	0.122 **	0.121 **
	(0.0596)	(0.0612)	(0.0612)	(0.0614)	(0.0612)
Independent directors	-0.550 ***	-0.564 ***	-0.554 ***	-0.524 ***	-0.519 ***
-	(0.167)	(0.173)	(0.175)	(0.175)	(0.176)
Largest shareholder	0.0366 ***	0.0349 ***	0.0350 ***	0.0376 ***	0.0387 ***
-	(0.0109)	(0.0111)	(0.0112)	(0.0117)	(0.0117)
SOE	0.829 **	0.816 **	0.823 **	0.737 *	0.788 **
	(0.368)	(0.396)	(0.402)	(0.381)	(0.382)
Leverage	-0.481	-0.610*	-0.637 *	-0.660 *	-0.619 *
-	(0.317)	(0.349)	(0.358)	(0.373)	(0.370)
Growth	-0.0175	-0.0176	-0.0276	-0.00720	-0.00905
	(0.0243)	(0.0249)	(0.0301)	(0.0274)	(0.0276)
Free cash flow	0.0604	0.0411	0.0421	0.0394	0.0409
	(0.0681)	(0.0691)	(0.0696)	(0.0700)	(0.0701)
SA	21.26 ***	20.61 ***	20.66 ***	21.00 ***	21.12 ***
	(2.008)	(2.049)	(2.061)	(2.110)	(2.108)
Log(GDP)		-0.249	0.127	-0.0324	-0.330
		(0.537)	(0.535)	(0.542)	(0.562)
Constant	54.05 ***	54.80 ***	50.04 ***	49.72 ***	57.87 ***
	(7.274)	(9.093)	(8.898)	(9.480)	(10.60)
Industry dummies	No	No	No	Yes	Yes
Province dummies	No	No	Yes	Yes	Yes
Year dummies	No	No	No	No	Yes
Firm dummies	Yes	Yes	Yes	Yes	Yes
Ν	4478	4186	4186	4186	4186
adj. R <sup>2</sup>	0.795	0.798	0.798	0.799	0.799

Table 5. Baseline results.

Standard errors in parentheses, \* p < 0.10, \*\* p < 0.05, \*\*\* p < 0.01. The standard errors are computed using robust standard errors.

Column (2) includes the log form of city-level GDP per capita to capture the macroeconomic impacts. Column (3) also includes province dummies to control for provincial fixed effects. In addition, industry dummies are added in column (4), and year dummies are further included in column (5). We use the specification of column (5) as our baseline model. The estimate of *Treat* \* *Post* is 1.574, indicating that the implementation of the green finance pilot zones policy has a significant and positive effect on firms' ESG performance. The result is consistent across all specifications. This result corroborates the studies of Li et al., Pang et al., and Su et al. [23,28,93], who found that green finance policy leads to firms' CSR increases and environmental performance.

Consistent with other ESG studies in China, our result is also economically nontrivial. Based on the estimations in column (5), the ESG score increases by 1.57, which is around 7.3% when evaluating at the sample mean of ESG. By examining listed firms in China during 2010–2019, Shu and Tan found that the carbon control policy reduces firm ESG performance by 0.011, which varies by 8.308% when the risk of carbon control policy varies by one standard deviation [16]. Fang et al. examined the impact of digitization on Bloomberg ESG using listed firms in China from 2012 to 2020. Their results suggested that the increase in digitalization could account for about 7.69% of the rise in ESG scores over the sample period.

The coefficients of other control variables are in line with our expectations. The results in column (5) show that larger firm size, older firms, the number of female directors, higher ownership concentration, SOE, and fewer financial constraints are all related to a higher ESG score. Leverage has a negative effect on ESG performance.

Equation (1) identifies the average effect of the green finance policy. Practically, the policy may have a time lag effect because the impact of the pilot zones policy might become more observable three or four years after the implementation. Moreover, the assumption of our DID identification strategy is that the different over-time changes between the treatment group and the control group should come solely from the policy implementation in 2017 and not from any preceding different time trend across firms. To test this assumption, in Equation (2) we replace the treat-post interaction in Equation (1) with the sum of the treatyear interactions as in previous studies [7,25,28,29,38]. The dynamic effects of the policy are presented in Table 6. As in Table 5, we also include different sets of industry, provincial and year dummies to control different fixed effects across different columns, with column (3) of Table 6 including the full set of fixed effects. As presented in column (3) of Table 6, all coefficients of interactions *treat* \* *year* are insignificant before and at the time of policy implementation, 2017, implying insignificant differences in pre-trends, suggesting that the parallel trend assumption of the DID method is not violated. The coefficients of interactions *treat* \* *year* are positive and statistically significant. The policy impact gradually increases after the initial policy year, consistent with our expectation.

The results in column (3) of Table 6 are graphed in Figure 2. The interval for each point indicates the 95% confidence interval. Clearly, there was no significant effect on firms' ESG performance before the implementation of the pilot zones policy in 2017. Nevertheless, there is a significant and positive effect on ESG after implementing the policy. Thus, similar to Table 6, Figure 3 also provides parallel trend tests, showing that the prerequisite of DID is satisfied.

### 5.2. The Effects of the Green Finance Policy on Individual ESG Pillars

We have seen that the green finance pilot zones policy has a positive and significant effect on firms' ESG activities in total. As the composite ESG score comprises three different pillars, it would be meaningful to examine if the pilot zones policy could also affect each ESG pillar. To answer these questions, we re-estimate the baseline model by replacing the total ESG score with the individual scores. The engagement activities across different ESG dimensions represent various aspects of stakeholder concerns [68]. Chan et al. examined listed firms in the US from 1992 to 2010, and their results showed that firms would make strategic trade-offs between different dimensions [13]. Specifically, financially constrained

firms might first reduce spending on charitable and innovative giving. Yuan et al. subdivided the five CSR dimensions into third-party CSR and stakeholder CSR [63]. Their results showed that more able CEOs invest more in stakeholder-related CSR dimensions. Thus, we consider different dimensions of ESG as fundamental tradeoffs among different stakeholders.

Table 6. Dynamic effects of the green finance policy.

\_\_\_\_

	(1)	(2)	(3)
Treat * year 2014	-0.204	-0.156	-0.182
-	(0.915)	(0.916)	(0.911)
Treat * year 2015	0.234	0.265	0.252
	(0.872)	(0.875)	(0.873)
Treat * year 2016	0.192	0.246	0.230
	(0.875)	(0.878)	(0.875)
Treat * year 2017	1.090	1.147	1.143
	(0.783)	(0.786)	(0.785)
Treat * year 2018	1.598 *	1.714 **	1.684 **
	(0.826)	(0.828)	(0.829)
Treat * year 2019	1.768 **	1.8/8 **	1.825 **
E (* 2000)	(0.854)	(0.854)	(0.855)
Ireat * year 2020	1.854 *	1.921 *	1.955 *
Einer eine	(0.996)	(0.994)	(1.009)
Firm size	(0.102)	(0.201)	(0.214)
ROA	(0.195)	(0.201)	(0.214)
KOA	0.514	0.524	0.567
Duality	0.049)	0.0845	(0.307) 0.137
Duanty	(0.207)	(0.209)	(0.211)
Female directors	0.0716	0.126 **	0 123 **
Tentale uncetors	(0.0597)	(0.0612)	(0.0614)
Independent directors	-0.547 ***	-0.557 ***	-0.519 ***
	(0.168)	(0.174)	(0.176)
Largest shareholder	0.0377 ***	0.0362 ***	0.0387 ***
0	(0.0109)	(0.0111)	(0.0117)
SOE	0.890 **	0.887 **	0.795 **
	(0.369)	(0.397)	(0.383)
Leverage	-0.450	-0.562	-0.619 *
	(0.318)	(0.347)	(0.370)
Growth	-0.0197	-0.0204	-0.00953
	(0.0246)	(0.0253)	(0.0276)
Free cash flow	0.0618	0.0439	0.0415
	(0.0683)	(0.0694)	(0.0702)
SA	21.41 ***	20.78 ***	21.15 ***
	(2.009)	(2.049)	(2.110)
Log(GDP)		-0.518	-0.303
Granting		(0.555)	(0.564)
Constant	(8.79***	82.23	//.16 ***
Industry dummics	(8.632) No	(10.54) No	(10.96)
Browince dummies	No	No	Tes
Vear dummies	No	No	Voc
Firm dummies	Yes	Yee	Yee
N	4478	4186	4186
adi $\mathbb{R}^2$	0 795	0 798	0 799
auj. n	0.755	0.770	0.777

Standard errors in parentheses, \* p < 0.10, \*\* p < 0.05, \*\*\* p < 0.01. The standard errors are computed using robust standard errors.



Figure 2. The dynamic impact of the green finance policy on the ESG performance.



Figure 3. Kernel densities pre and post-matching (Nearest-neighbor matching with 0.05 caliper).

As can be seen from Table 7, the overall positive effect of the pilot zones policy on ESG performance is driven by the environmental pillar. For the environmental pillar in column (1), the coefficient of interaction term *treat* \* *year* is 1.764 and significant at the 5% level, implying that firms affected by the green finance pilot zones policy significantly enhance their ESG engagement compared to those not influenced by the policy.

	E	S	G
	(1)	(2)	(3)
Treat * post	1.764 **	0.918	0.314
-	(0.848)	(0.559)	(0.263)
Constant	83.89 ***	23.12	60.89 ***
	(15.63)	(14.34)	(8.722)
Control variables	Yes	Yes	Yes
Industry dummies	Yes	Yes	Yes
Province dummies	Yes	Yes	Yes
Year dummies	Yes	Yes	Yes
Firm dummies	Yes	Yes	Yes
N	3776	4173	4186
adj. R <sup>2</sup>	0.732	0.771	0.809

Table 7. The effects of green finance policy on individual ESG pillars.

Standard errors in parentheses, \* p < 0.10, \*\* p < 0.05, \*\*\* p < 0.01. The standard errors are computed using robust standard errors.

Notably, the coefficients of the interaction are insignificant in columns (2) and (3), indicating that the green finance policy fails to have a significant impact on the social and governance pillars. This is easy to understand since the green finance policy is designed to be effective in environmental aspects. For example, Zhang et al. examined the same green finance pilot policy, and they showed that the green finance policy is effective in air pollution control, which is an essential aspect of the environmental pillar of ESG [7]. Using provincial panel data during 2011–2019, Huang and Zhang examined the same policy, and they found that macroeconomically, the green finance pilot policy plays a vital role in pollution control and is conducive to environmental enhancement.

#### 5.3. Heterogeneity

The average effect of the green finance policy on firms' ESG performance was demonstrated above. Our sample covers various firms, and there are also differences in firms' ESG performance within industries. It is interesting, therefore, to explore the asymmetric effects of the policy. Yu et al. found that generally, green finance policies could effectively mitigate financing restraints on green innovation, but POEs are less likely to obtain green credits. Their results imply that POEs are in a disadvantaged position to get credits under the existing green finance system [4]. Moreover, Hong et al. showed that firms with fewer financial constraints are more likely to invest in CSR [18]. In addition, Xu and Li showed that the green credit policy has asymmetric impacts on "two high" firms and green firms [78]. Specifically, the green credit policy increases the debt financing cost of "two-high" firms but decreases that of green firms. Thus, as in other green finance policy studies [16,20,28,38,94], we also discussed the heterogeneity in this section. We use both subsample regression [16,20,94] and triple differences methods [28,38].

Table 8 shows the heterogenous effects of the pilot policy for non-SOEs versus SOEs, high financial constrained versus low financial constrained, and non-heavy-polluting versus heavy-polluting firms in columns (1)–(6), respectively.

In China, tighter connection with the government possibly grants SOEs better access to external financing, while this may not be the case for non-SOEs [38]. Therefore, we predict stronger impacts of the pilot policy on SOEs. As presented in columns (1) and (2) of Table 8, consistent with Zhang et al., we also find that the policy has a larger effect among state-owned firms [92]. Compared to SOEs, non-SOEs tend to face higher financing constraints due to uneven loan availability from state-owned banks [95]. If financial distress accounts for firms' poor environmental performance, the inferior environmental performance of non-SOEs implies that non-SOEs have limited access to financing resources. Jin et al. showed that banks are still the main provider of credit in the Chinese financial market [96]. This result is also consistent with the literature on the relationship between financial constraints and pollution control. Zhang et al. found that SOEs mainly resort to external financing

resources to reduce emissions, while POEs mainly financed internally [97]. To further examine the heterogenous effect of SOEs, we substitute the triple interaction in equation (3) with the triple interaction term *treat* \* *post* \* *SOE*. As shown in column (4) of Table 9, the effect of green finance on ESG performance is significantly positive for SOEs relative to non-SOEs.

indic of includencity.	Tabl	e 8.	Hetero	geneity.
------------------------	------	------	--------	----------

	Non-SOEs	SOEs	High Financial Constrained Firms	Low Financial Constrained Firms	Non-Heavy- Polluting Firms	Heavy-Polluting Firms
	(1)	(2)	(3)	(4)	(5)	(6)
Treat * post	1.106 **	2.255 ***	0.654	3.356 ***	1.403 **	1.299 ***
_	(0.438)	(0.682)	(0.784)	(0.814)	(0.680)	(0.491)
Constant	58.21 ***	38.44 **	20.90	86.67 ***	81.06 ***	26.34 *
	(15.05)	(14.95)	(17.17)	(21.01)	(14.15)	(15.09)
Control variables	Yes	Yes	Yes	Yes	Yes	Yes
Industry dummies	Yes	Yes	Yes	Yes	Yes	Yes
Province dummies	Yes	Yes	Yes	Yes	Yes	Yes
Year dummies	Yes	Yes	Yes	Yes	Yes	Yes
Firm dummies	Yes	Yes	Yes	Yes	Yes	Yes
Ν	2276	1895	2067	2061	2123	2056
adj. R <sup>2</sup>	0.809	0.784	0.812	0.816	0.817	0.779

Standard errors in parentheses, \* p < 0.10, \*\* p < 0.05, \*\*\* p < 0.01. The standard errors are computed using robust standard errors.

#### Table 9. Triple difference test of Heterogenous effects.

	(1)	(2)	(3)	(4)
Treat * post * Pollution	-0.00844			
	(0.839)			
Treat * post * highGDP		1.701 **		
		(0.819)		
Treat * post * highSA			2.746 ***	
1 0			(1.042)	
Treat * post * SOE				1.698 **
				(0.785)
Constant	76.08 ***	74.14 ***	72.85 ***	75.20 ***
	(10.79)	(10.94)	(10.99)	(11.01)
Control variables	Yes	Yes	Yes	Yes
Industry dummies	Yes	Yes	Yes	Yes
Province dummies	Yes	Yes	Yes	Yes
Year dummies	Yes	Yes	Yes	Yes
Firm dummies	Yes	Yes	Yes	Yes
Ν	4186	4161	4186	4186
adj. R <sup>2</sup>	0.799	0.799	0.800	0.799

Standard errors in parentheses, \* p < 0.10, \*\* p < 0.05, \*\*\* p < 0.01. The standard errors are computed using robust standard errors.

Financial constraint serves as a key mechanism through which green finance influences firms' environmental performance. Thus, the green finance policy might have heterogenous effects on firms with different levels of financial constraints. The results in columns (3) and (4) of Table 8 show that the green finance policy has a more significant effect on firms with low financial constraints. Meanwhile, the green finance policy has no significant effect on

firms with high financial constraints. This implies that firms in a weak financial situation are less likely to engage in CSR. Consistent with our findings, Chan et al. suggested a significant negative relationship between financial constraints and CSR engagement [13]. To further examine the heterogenous effect of financial constraints, we substitute the triple interaction in Equation (3) with the triple interaction term *treat* \* *post* \* *highSA*, where *highSA* is a dummy variable equal to 1 for firms with SA index above the sample median (low financial constraints). As shown in column (3) of Table 9, the effect of green finance on ESG performance is significantly positive for less financially constrained firms than firms with high financial constraints.

Environmental damage is mainly caused by heavy-polluting firms. Thus, it is important to assess whether green finance policy could promote ESG activities of heavy-polluting firms. Results in columns (5) and (6) of Table 8 show that the green finance policy plays an important role in both non-heavy-polluting and heavy-polluting firms. To further investigate the effect of green finance policy on firms' ESG performance varied by pollution level, we use a triple interaction term as specified in Equation (3), as in previous literature [38]. As shown in column (1) of Table 9, the effect of the green finance policy on firms' ESG performance is insignificant for heavy-polluting firms relative to non-heavy-polluting firms, indicating that there's no significant heterogeneity between heavy-polluting and non-heavy-polluting firms.

As indicated in Table 1, the green finance pilot zones policy has different priorities in different regions. Financial support is largely determined by the level of local economic development. For example, in Zhejiang and Jiangsu, which are more economically developed, more advanced instruments, such as green bonds and green credit systems, are stressed. In Xinjiang, the priority is to optimize local ecological resources. We use city-level GDP per capita to measure the level of economic development. To examine the effect of heterogenous green finance policy in different pilot zones, we substitute the triple interaction term in Equation (3) with the triple interaction term *treat* \* *post* \* *highGDP*, where *highGDP* is a dummy variable equal to 1 for firms in pilot zones with city-level GDP per capita above the sample median and 0 otherwise. As shown in column (2) of Table 9, the effect of green finance policy on firms' ESG performance is significantly positive for more developed zones relative to less developed areas. Consequently, the ESG performance of firms in more developed zones improves more when exposed to the pilot policy.

#### 5.4. Mechanism Analysis

Based on the above analysis, we have shown that the green finance pilot policy has a positive and significant effect on firms' ESG engagement. As we mentioned before, previous studies have pointed out that financial constraints are a key driver of corporate social responsibility [18,36]. To show the role of financial constraints in the relationship between the green finance pilot zones policy and firms' ESG performance, we substitute the dependent variable with the SA index and re-estimate Equation (1).

Column (1) of Table 10 presents the results. Consistent with Li et al., the pilot policy has a negative and significant effect on the SA index, indicating that the pilot policy tightens firms' financial constraints [28]. They investigated the 2012 China's Green Credit Guidelines using listed firms from 2009–2020, and they found that the green credit policy could worsen firms' financial constraints. Moreover, our result is consistent with Liu et al., which utilized a financial computable general equilibrium model (CGE) and found that with the implementation of the green credit policy, the average financing costs of the energy-intensive industries increase by 1.1% in the short run, decrease by 0.008% in the medium run, and increase by 0.003% in the long run [77]. The pilot zones policy was implemented in June 2017, and our sample period is 2013–2020. Thus, we might only observe the short-run effect of the policy. The higher financing costs further tighten financial constraints in the short run. Despite the differences in priorities in different pilot zones, the pilot zones all use capital allocation to steer financial resources into green projects and technologies.

	Full Sample	High FC	Low FC	Non-SOEs	SOEs	Non-Heavy- Polluting Firms	Heavy-Polluting Firms
Treat * post	(1) -0.0140 *** (0.00498)	(2) 0.00495 (0.00719)	(3) -0.0150 (0.0112)	(4) -0.0315 *** (0.00588)	(5) -0.000647 (0.00774)	(6) -0.00517 (0.00687)	(7) -0.0183 *** (0.00691)
Constant	-3.734 *** (0.0788)	-3.966 *** (0.0731)	-3.626 *** (0.188)	-3.355 *** (0.116)	-3.864 *** (0.105)	-3.568 *** (0.130)	-3.746 *** (0.0975)
Control variables	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Industry dummies	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Province dummies	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Year dummies	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Firm dummies	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Ν	4186	2067	2061	2276	1895	2123	2056
adj. R <sup>2</sup>	0.975	0.968	0.932	0.976	0.974	0.978	0.970

Thus, heavy-polluting firms could easily suffer from higher thresholds and capital costs in external financing [37].

**Table 10.** Mechanism analysis: financial constraints.

Standard errors in parentheses, \* p < 0.10, \*\* p < 0.05, \*\*\* p < 0.01. The standard errors are computed using robust standard errors. Firm age and firm size are not included as control variables.

To further analyze the heterogenous effect on financial constraints, we re-estimate the same specification as in column (1) of Table 10 but using the subsamples: firms with high financial constraints (with the SA index lower than the sample median), firms with low financial constraints, SOEs, non-SOEs, non-heavy-polluting firms, and heavy-polluting firms for columns (2)–(7), respectively. The results show that the pilot policy tightens financial constraints significantly in non-SOEs and heavy-polluting firms but has no significant effect in both highly and low-constrained firms, SOEs, and non-heavy-polluting firms.

Unsurprisingly, the heavy-polluting firms become more financially constrained since the main aim of the green finance policy is to support environment-friendly projects. Our results indicate that the pilot policy has worsened the financial constraints of less constrained firms more relative to high-constrained firms, though insignificantly. This might be due to the sample composition. For example, there are 56.1% of non-SOEs in less constrained firms, while 53.4% in highly constrained firms. Non-SOEs are more likely to be financially constrained because of the pilot policy, as shown in columns (4) and (5). Yu et al. showed that green finance policies primarily mitigated the financing constraints of SOEs, and POEs are less likely to obtain green credits [4].

In sum, the mechanism analysis reveals that the pilot policy promotes firms' ESG performance even if it worsens their overall financial constraints. The reason might be that the policy aims at environment-friendly projects. Moreover, the effects on financial constraints are heterogenous, with less financially constrained firms, non-SOEs, and heavy-polluting firms experiencing a statistically significant increase in financial constraints when exposed to the pilot zones policy.

## 6. Robustness Tests

## 6.1. PSM-DID

The pilot policy is implemented in only eight cities, and therefore only a relatively small share of firms are exposed to the policy. Thus, there may exist a small sample selection bias [37]. To mitigate the potential self-selection bias, as in other studies [16,20,25,28,38], we utilize the propensity score matching (PSM) method to match firms in pilot cities and firms not exposed to the policy. The key idea of PSM is to construct a counterfactual control group to investigate the effect of the changes on ESG performance. We use one-to-three nearest-neighbor matching with 0.05 caliper, one-to-three nearest-neighbor matching

with 0.001 caliper, radius matching with 0.001 radius, and kernel matching to show the consistency of our results.

To demonstrate the difference in pre and post-matching, Figure 3 presents the kernel densities of scores for the two groups. The left part depicts kernel densities for the treatment and control groups before matching, while the right part presents kernel densities post-matching (one-to-three nearest-neighbor matching with 0.05 caliper). The distributions after matching are more consistent and almost overlap with each other.

Table 11 shows the PSM-DID results. They are quite similar to the baseline regression findings. That is, the net effect of the green finance policy on firms' ESG performance is still positive and significant, implying that our baseline results are not biased by unobserved factors.

	Nearest (c = 0.05)	Nearest (c = 0.001)	Radius	Kernel
	(1)	(2)	(3)	(4)
Treat * post	1.601 ***	1.667 ***	1.667 ***	1.601 ***
	(0.396)	(0.403)	(0.403)	(0.396)
Constant	62.69 ***	59.46 ***	59.46 ***	62.69 ***
	(11.88)	(11.98)	(11.98)	(11.88)
Control variables	Yes	Yes	Yes	Yes
Industry dummies	Yes	Yes	Yes	Yes
Province dummies	Yes	Yes	Yes	Yes
Year dummies	Yes	Yes	Yes	Yes
Firm dummies	Yes	Yes	Yes	Yes
Ν	3920	3761	3761	3920
adj. R <sup>2</sup>	0.798	0.801	0.801	0.798

Table 11. PSM-DID results.

Standard errors in parentheses, \* p < 0.10, \*\* p < 0.05, \*\*\* p < 0.01. The standard errors are computed using robust standard errors. The propensity score is estimated using logit model. Province dummies, industry dummies, and year dummies are controlled.

## 6.2. Alternative Proxy for ESG Performance

Similar to other ESG studies, the ESG scores might suffer from reporting bias. Only about 30% of listed firms have Bloomberg ESG ratings, due to inadequate ESG disclosure information [85]. Moreover, Sun and Saat mentioned that Bloomberg only covers about 1000 listed firms in China [61]. To rule out the possibility that the coverage of the ESG score might bias the results, we also use Runling CSR to show the robustness of our results. Runling ESG is retrieved from the CSR rating agency Runling (also known as RKS). RKS organization and the Runling CSR scores for Chinese firms are modeled after the US CSR rating agency KLD. Thus, similar to studies based on KLD CSR scores [52,98], the Runling CSR score is also widely used as a composite measure of CSR for Chinese firms in the literature [99–102]. Table 12 shows the results using Runling CSR ratings as the alternative proxy for ESG performance. The results presented in Table 12 are consistent with the baseline results.

# 6.3. Placebo Test

To examine if omitted variables bias the results, as in other studies [6,7,15,19,97], a placebo test is performed by randomly assigning the implementation of the pilot policy to cities [103]. Eight cities are randomly selected at time *t*, which are selected at random between 2014 and 2018, allowing for at least one year before and one year after the policy to implement the DID method. Using this false green finance status variable and time variable, we conduct a placebo estimation using the same specification in column (5) of Table 5. The false green finance policy variable should have generated an insignificant estimate with a

magnitude near zero; otherwise, it would suggest a misspecification of the DID estimation. It is repeated 500 times to enhance the identification power of this placebo test.

	(1)	(2)	(3)	(4)
Treat * post	2.708 **	2.709 **	2.826 **	2.894 **
-	(1.267)	(1.267)	(1.328)	(1.309)
Constant	-12.38	-13.39	-13.28	-3.926
	(19.71)	(19.79)	(19.87)	(20.81)
Control variables	Yes	Yes	Yes	Yes
Industry dummies	No	No	Yes	Yes
Province dummies	No	Yes	Yes	Yes
Year dummies	No	No	No	Yes
Firm dummies	Yes	Yes	Yes	Yes
N	2060	2060	2060	2060
adj. R <sup>2</sup>	0.834	0.833	0.833	0.837

Standard errors in parentheses, \* p < 0.10, \*\* p < 0.05, \*\*\* p < 0.01. The standard errors are computed using robust standard errors.

Figure 4 presents the distribution of the estimates from the 500 runs. We can see that the observations are scattered and resemble a symmetric bell-like shape. Moreover, the *p*-value of the Jarque–Bera test of the estimates is 0.1694. Thus, we fail to reject the null hypothesis of the normality of the distributions of the estimates. The estimates are normally distributed and scattered around 0, indicating that an artificial "counterfeit variable" has no effect on firms' ESG performance. The results imply that it is the green finance pilot zones policy rather than other policies in pilot zones that have led to the increase in firms' ESG performance.



**Figure 4.** Distribution of estimated coefficients of falsification test. The figure shows the cumulative distribution density of the estimated coefficients is from 500 simulations randomly assigning the green finance policy status to cities.

### 6.4. Sustainability Implications and Policy Recommendations

Our results have important sustainability implications. Firstly, green finance policy could be a powerful instrument to enhance firms' sustainability. Green finance is designed to make sustainability a part of financial decision making. Our results indicate that green finance policy is effective in fostering firms' sustainability, measured by ESG ratings. Moreover, better ESG performance is usually related to higher firm value, less risk, and better access to finance [54–57].

Secondly, better ESG performance signals firms' endeavor in reducing pollution and improving energy efficiency. In particular, our results indicate that firms' ESG growth has been mainly driven by the environmental pillar. Thus, the improvement in firms' ESG performance induced by green finance policy is vital for environmental protection, climate change mitigation, and thus for society's sustainability.

Thirdly, different from other studies, our results imply that green finance policy could promote sustainability through sustainable investment. Socially responsible investors rely heavily on the ESG scores [11]. However, ESG investing has yet to become mainstream in China. Globally, around 33% of all professionally managed assets are subject to ESG criteria [104]. The number of global investor signatories to the United Nations Principles for Responsible Investment (UN PRI) topped 3000 in 2020, up from 63 in 2006. Total assets under management (AUM) represented by the signatories exceeded USD 100 trillion in 2020 [105]. According to PwC's prediction, ESG-oriented AUM in the US would increase to 10.5 trillion dollars in 2026 from 4.5 trillion US dollars in 2021; in Europe, it would grow 53% to 19.6 trillion dollars. The Asia-Pacific is expected to more than triple, reaching 3.3 trillion dollars [106]. In China, except for money market funds, total ESG exchange-traded funds (ETF) and mutual funds experienced AUM more than triple in 2020 to reach 28.5 billion dollars, with net inflows reaching 10.5 billion dollars [107].

Our results indicate that the green finance policy plays a vital role in fostering firms' ESG, which is the basis of sustainable investing. Thus, our results imply that the green finance policy could be supportive of sustainable investment in China. Pástor et al. suggested that sustainable investing exerts its social impacts in two ways [108]. First, it promotes firms to become greener. Second, it encourages green firms to invest more and decreases investment by brown firms. Therefore, our results also have a vital implication for developing a greener society at large.

The results have important policy implications. ESG increases a firm's value and also serves as a vital driving force to promote the sustainable development of firms. However, motivating firms to engage more in ESG in emerging countries like China is still challenging. Our results imply that green finance could be an effective instrument to improve firms' ESG performance. Moreover, while vigorously promoting green finance, the policymaker should also allocate more financial resources to non-SOEs which are usually less financially constrained.

## 7. Conclusions

This paper explores the 2017 green finance pilot zones policy as a quasi-natural experiment to examine the effects of the green finance policy on firms' ESG performance. The findings show that, firstly, the 2017 green finance pilot zones policy has had a significant and positive effect on firms' ESG performances. Secondly, the overall positive effect of the green finance policy on ESG performance is driven mainly by the environmental pillar rather than the social and governance pillars.

Thirdly, utilizing the subsample estimation and triple differences method, we further find that the higher ESG performance is driven mainly by heavy-polluting firms, firms with less financial constraints, firms in economically more developed zones, and SOEs. Fourthly, the mechanism analysis reveals that the pilot policy improves firms' ESG performance even if it worsens their overall financial constraints. The reason might be that the policy aims at environment-friendly projects. Moreover, the effects on financial constraints are heterogenous, with less financially constrained firms, non-SOEs, and heavy-polluting firms experiencing a statistically significant increase in financial constraints when exposed to the pilot zones policy. Finally, the results are robust to the parallel trend test, PSM-DID, alternative ESG proxy, and placebo test.

Previous studies have shown that green finance policy could foster firms' green performance and green innovation [20,23]. Consistent with their findings, our results suggest that green finance policy could enhance the environmental pillar of a firm's ESG. Moreover, different from their conclusions, we find that the green finance policy has a positive and significant effect on firms' ESG performance. Socially responsible investors rely heavily on the ESG scores [11]. Therefore, our results have important implications for establishing a better green finance system in China.

Moreover, heterogeneity analysis shows that the higher ESG performance is mainly driven by firms with less financial constraints, firms in economically more developed areas, and SOEs. This confirms that firms with fewer financial constraints are more likely to engage in ESG [13,18].

In addition, consistent with Li et al. [28], the mechanism analysis reveals that the pilot policy fosters firms' ESG performance even if it worsens their overall financial constraints. The reason might be that the policy aims at environment-friendly projects. Moreover, our results show that the tightening financial impacts are asymmetric among different types of firms. Specifically, the green finance pilot policy worsens the financial constraints of non-SOE firms and high-polluting firms more. Our results confirm the findings of Yu et al. and Xu and Li [4,78].

However, this paper has the following limitations. Firstly, due to insufficient ESG disclosure information, our sample only includes listed firms, which usually outperform non-listed firms in size and profitability. Thus, our conclusion cannot be directly extrapolated to non-listed firms. In the future, when there are more data on ESG performance, the research could extend to a wider range of firms. Secondly, our results indicate that the pilot zones policy worsens firms' overall financial constraints. In the literature, Li et al. found that green finance policy increases firms' overall financial constraints [28], but Yu et al. showed that green finance policies could effectively ease financial constraints on green finance [4]. The reason might be that the pilot zones policy was first implemented in June 2017, and our sample period is 2013–2020. Thus, we might only be observing the short-run effect of the policy. We look forward to further research with an expanded sample period to examine the long-term effect of the pilot zones policy on firms' financial constraints. Thirdly, previous studies have shown that financial constraint is an essential channel through which green finance impacts firms' ESG performance. However, as we have no access to detailed firm-level or regional-level green credit data, we could not examine the underlying mechanisms in depth. In the future, more detailed information on green credit, green insurance, green bonds, and green security could help us to dig deep into the underlying mechanisms of the green finance policy, and the corresponding conclusions should help us to better understand the effects of green finance.

**Author Contributions:** Conceptualization, X.S. and Z.G.; methodology, X.S. and Z.G.; software, X.S. and C.Z.; validation, X.S., C.Z. and Z.G.; formal analysis, X.S.; investigation, X.S., C.Z. and Z.G.; resources, X.S.; data curation, X.S.; writing—original draft preparation, X.S.; writing—review and editing, Z.G.; visualization, C.Z.; supervision, X.S.; project administration, X.S.; funding acquisition, X.S. All authors have read and agreed to the published version of the manuscript.

**Funding:** This research was funded by the National Natural Science Foundation of China (Grant #72103167).

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: Not applicable.

**Acknowledgments:** We would like to thank the editor and anonymous referees for their constructive comments that helped to improve the paper.

# Conflicts of Interest: The authors declare no conflict of interest.

# References

- 1. Nordhaus, W. Climate Change: The Ultimate Challenge for Economics. Am. Econ. Rev. 2019, 109, 1991–2014. [CrossRef]
- Hou, H.; Zhu, Y.; Wang, J.; Zhang, M. Will Green Financial Policy Help Improve China's Environmental Quality? The Role of Digital Finance and Green Technology Innovation. *Environ. Sci. Pollut. Res.* 2023, 30, 10527–10539. [CrossRef] [PubMed]
- The World Bank. China's Transition to a Low-Carbon Economy and Climate Resilience Needs Shifts in Resources and Technologies. 2022. Available online: https://www.worldbank.org/en/news/press-release/2022/10/12/china-s-transition-to-a-low-carboneconomy-and-climate-resilience-needs-shifts-in-resources-and-technologies (accessed on 12 October 2022).
- 4. Yu, C.H.; Wu, X.; Zhang, D.; Chen, S.; Zhao, J. Demand for Green Finance: Resolving Financing Constraints on Green Innovation in China. *Energy Policy* **2021**, *153*, 112255. [CrossRef]
- The World Economic Forum. Making Net-Zero Happen: How China Can Finance Its Transition. 2022. Available online: https://www.weforum.org/agenda/2022/07/making-net-zero-happen-how-china-can-finance-its-transition/ (accessed on 21 July 2022).
- Sartzetakis, E.S. Green Bonds as an Instrument to Finance Low Carbon Transition. *Econ. Chang. Restruct.* 2021, 54, 755–779. [CrossRef]
- 7. Zhang, A.; Wang, S.; Liu, B. How to Control Air Pollution with Economic Means? Exploration of China's Green Finance Policy. *J. Clean. Prod.* **2022**, *353*, 131664. [CrossRef]
- 8. Irfan, M.; Razzaq, A.; Sharif, A.; Yang, X. Influence Mechanism between Green Finance and Green Innovation: Exploring Regional Policy Intervention Effects in China. *Technol. Forecast. Soc. Chang.* **2022**, *182*, 121882. [CrossRef]
- 9. Green Finance Task Force. Establishing China's Green Financial System; People's Bank China United Nations Environment Programme: 2015. Available online: https://www.unep.org/resources/report/establishing-chinas-green-financial-system (accessed on 9 April 2023).
- 10. Patel, P.C.; Pearce, J.A.; Oghazi, P. Not so Myopic: Investors Lowering Short-Term Growth Expectations under High Industry ESG-Sales-Related Dynamism and Predictability. *J. Bus. Res.* **2021**, *128*, 551–563. [CrossRef]
- 11. Drempetic, S.; Klein, C.; Zwergel, B. The Influence of Firm Size on the ESG Score: Corporate Sustainability Ratings Under Review. *J. Bus. Ethics* **2020**, *167*, 333–360. [CrossRef]
- 12. Li, T.T.; Wang, K.; Sueyoshi, T.; Wang, D.D. ESG: Research Progress and Future Prospects. Sustainability 2021, 13, 11663. [CrossRef]
- 13. Chan, C.Y.; Chou, D.W.; Lo, H.C. Do Financial Constraints Matter When Firms Engage in CSR? *N. Am. J. Econ. Financ.* **2017**, *39*, 241–259. [CrossRef]
- 14. Baldini, M.; Maso, L.D.; Liberatore, G.; Mazzi, F.; Terzani, S. Role of Country- and Firm-Level Determinants in Environmental, Social, and Governance Disclosure. *J. Bus. Ethics* **2018**, *150*, 79–98. [CrossRef]
- 15. DasGupta, R. Financial Performance Shortfall, ESG Controversies, and ESG Performance: Evidence from Firms around the World. *Financ. Res. Lett.* **2022**, *46*, 102487. [CrossRef]
- 16. Shu, H.; Tan, W. Does Carbon Control Policy Risk Affect Corporate ESG Performance? Econ. Model. 2023, 120, 106148. [CrossRef]
- 17. Campbell, J.L. Why Would Corporations Behave in Socially Responsible Ways? An Institutional Theory of Corporate Social Responsibility. *Acad. Manag. Rev.* 2007, 32, 946–967. [CrossRef]
- 18. Hong, H.; Kubik, J.D.; Scheinkman, J.A. Financial Constraints on Corporate Goodness. In *NBER Working Paper No. 18476*; 2012. Available online: https://www.nber.org/system/files/working\_papers/w18476/w18476.pdf (accessed on 27 February 2023).
- 19. Ren, X.; Shao, Q.; Zhong, R. Nexus between Green Finance, Non-Fossil Energy Use, and Carbon Intensity: Empirical Evidence from China Based on a Vector Error Correction Model. *J. Clean. Prod.* **2020**, 277, 122844. [CrossRef]
- Liu, S.; Wang, Y. Green Innovation Effect of Pilot Zones for Green Finance Reform: Evidence of Quasi Natural Experiment. *Technol. Forecast. Soc. Change* 2023, 186, 122079. [CrossRef]
- 21. Chen, X.; Chen, Z. Can Green Finance Development Reduce Carbon Emissions? Empirical Evidence from 30 Chinese Provinces. *Sustainability* **2021**, *13*, 12137. [CrossRef]
- 22. Zhang, B.; Wang, Y. The Effect of Green Finance on Energy Sustainable Development: A Case Study in China. *Emerg. Mark. Financ. Trade* **2021**, *57*, 3435–3454. [CrossRef]
- 23. Su, Z.; Guo, Q.; Lee, H.-T. Green Finance Policy and Enterprise Energy Consumption Intensity: Evidence from a Quasi-Natural Experiment in China. *Energy Econ.* **2022**, *115*, 106374. [CrossRef]
- 24. Yan, C.; Mao, Z.; Ho, K.C. Effect of Green Financial Reform and Innovation Pilot Zones on Corporate Investment Efficiency. *Energy Econ.* **2022**, *113*, 106185. [CrossRef]
- 25. Hu, G.; Wang, X.; Wang, Y. Can the Green Credit Policy Stimulate Green Innovation in Heavily Polluting Enterprises? Evidence from a Quasi-Natural Experiment in China. *Energy Econ.* **2021**, *98*, 105134. [CrossRef]
- Shi, J.; Yu, C.; Li, Y.; Wang, T. Does Green Financial Policy Affect Debt-Financing Cost of Heavy-Polluting Enterprises? An Empirical Evidence Based on Chinese Pilot Zones for Green Finance Reform and Innovations. *Technol. Forecast. Soc. Change* 2022, 179, 121678. [CrossRef]
- 27. Huang, H.; Zhang, J. Research on the Environmental Effect of Green Finance Policy Based on the Analysis of Pilot Zones for Green Finance Reform and Innovations. *Sustainability* **2021**, *13*, 3754. [CrossRef]

- Li, S.; Liu, Q.; Lu, L.; Zheng, K. Green Policy and Corporate Social Responsibility: Empirical Analysis of the Green Credit Guidelines in China. J. Asian Econ. 2022, 82, 101531. [CrossRef]
- 29. Christensen, D.M.; Serafeim, G.; Sikochi, A. Why Is Corporate Virtue in the Eye of The Beholder? The Case of ESG Ratings. *Account. Rev.* 2022, 97, 147–175. [CrossRef]
- Peng, H.; Lu, Y.; Gupta, S.; Wang, Q. Dynamic and Heterogeneity Assessment of Carbon Efficiency in the Manufacturing Industry in China: Implications for Formulating Carbon Policies. *Environ. Impact Assess. Rev.* 2022, 97, 106885. [CrossRef]
- Wang, H.; Li, B. Environmental Regulations, Capacity Utilization, and High-Quality Development of Manufacturing: An Analysis Based on Chinese Provincial Panel Data. Sci. Rep. 2021, 11, 1–13. [CrossRef]
- 32. D'Orazio, P.; Valente, M. The Role of Finance in Environmental Innovation Diffusion: An Evolutionary Modeling Approach. J. Econ. Behav. Organ. 2019, 162, 417–439. [CrossRef]
- Madaleno, M.; Dogan, E.; Taskin, D. A Step Forward on Sustainability: The Nexus of Environmental Responsibility, Green Technology, Clean Energy and Green Finance. *Energy Econ.* 2022, 109, 105945. [CrossRef]
- Wang, Q.J.; Wang, H.J.; Chang, C.P. Environmental Performance, Green Finance and Green Innovation: What's the Long-Run Relationships among Variables? *Energy Econ.* 2022, 110, 106004. [CrossRef]
- Leong, C.K.; Yang, Y.C. Constraints on "Doing Good": Financial Constraints and Corporate Social Responsibility. *Financ. Res.* Lett. 2021, 40, 101694. [CrossRef]
- 36. Xu, Q.; Kim, T. Financial Constraints and Corporate Environmental Policies. Rev. Financ. Stud. 2022, 35, 576–635. [CrossRef]
- Huang, H.; Mbanyele, W.; Wang, F.; Song, M.; Wang, Y. Climbing the Quality Ladder of Green Innovation: Does Green Finance Matter? *Technol. Forecast. Soc. Change* 2022, 184, 122007. [CrossRef]
- 38. Fan, H.; Peng, Y.; Wang, H.; Xu, Z. Greening through Finance? J. Dev. Econ. 2021, 152, 102683. [CrossRef]
- 39. Zhou, B.; Zhang, C. When Green Finance Meets Banking Competition: Evidence from Hard-to-Abate Enterprises of China. *Pacific-Basin Financ. J.* **2023**, *78*, 101954. [CrossRef]
- 40. Li, Q.; Zhang, K.; Wang, L. Where's the Green Bond Premium? Evidence from China. *Financ. Res. Lett.* 2022, 48, 102950. [CrossRef]
- MSCI. China's Green-Bond Market: Growing Issuance and Historical Outperformance. 2022. Available online: https://www. msci.com/www/blog-posts/china-s-green-bond-market/03029315810 (accessed on 24 February 2022).
- Green Central Banking. Uncovering China's Green Finance Pilot Zones. 2022. Available online: https://greencentralbanking. com/2022/06/01/china-green-finance-pilot-zones-mengwei-sha/ (accessed on 1 June 2022).
- 43. Paulson Institute. A Progress Report on China's Green Finance Pilot Zones. 2020. Available online: https://www.paulsoninstitute. org/green-finance/green-scene/a-progress-report-on-chinas-green-finance-pilot-zones/ (accessed on 17 October 2020).
- 44. Lv, C.; Bian, B.; Lee, C.C.; He, Z. Regional Gap and the Trend of Green Finance Development in China. *Energy Econ.* **2021**, 102, 105476. [CrossRef]
- 45. Freeman, R.E.; Wicks, A.C.; Parmar, B. Stakeholder Theory and "The Corporate Objective Revisited". *Organ. Sci.* **2004**, *15*, 364–369. [CrossRef]
- Deng, X.; Kang, J.k.; Low, B.S. Corporate Social Responsibility and Stakeholder Value Maximization: Evidence from Mergers. J. Financ. Econ. 2013, 110, 87–109. [CrossRef]
- 47. Margolis, J.D.; Walsh, J.R. Misery Loves Rethinking Companies: Social Initiatives. Adm. Sci. Quaterly 2003, 48, 268–305. [CrossRef]
- 48. Garriga, E.; Mele, D. Corporate Social Responsibility Theories: Mapping the Territory. J. Bus. Ethics 2004, 53, 51–71. [CrossRef]
- Hasan, I.; Kobeissi, N.; Liu, L.; Wang, H. Corporate Social Responsibility and Firm Financial Performance: The Mediating Role of Productivity. J. Bus. Ethics 2018, 149, 671–688. [CrossRef]
- Tang, Z.; Hull, C.E.; Rothenberg, S. How Corporate Social Responsibility Engagement Strategy Moderates the CSR-Financial Performance Relationship. J. Manag. Stud. 2012, 49, 1274–1303. [CrossRef]
- Hull, C.E.; Rothenberg, S. Firm Performance: The Interactions of Corporate Social Performance with Innovation and Industry Differentiation. *Strateg. Manag. J.* 2008, 29, 781–789. [CrossRef]
- 52. Awaysheh, A.; Heron, R.A.; Perry, T.; Wilson, J.I. On the Relation between Corporate Social Responsibility and Financial Performance. *Strateg. Manag. J.* 2020, *41*, 965–987. [CrossRef]
- 53. Cho, S.J.; Chung, C.Y.; Young, J. Study on the Relationship between CSR and Financial Performance. *Sustainability* **2019**, *11*, 343. [CrossRef]
- 54. Behl, A.; Kumari, P.S.R.; Makhija, H.; Sharma, D. Exploring the Relationship of ESG Score and Firm Value Using Cross-Lagged Panel Analyses: Case of the Indian Energy Sector. *Ann. Oper. Res.* **2022**, *313*, 231–256. [CrossRef]
- 55. Godfrey, P.C.; Merrill, C.B.; Hansen, J.M. The Relationship Between Corporate Social Responsibility and Shareholder Value: An Empirical Test of the Risk Management Hypothesis. *Strateg. Manag. J.* **2009**, *30*, 425–445. [CrossRef]
- Shiu, Y.-M.; Yang, S.-L. Does Engagement in Corporate Social Responsibility Provide Strategic Insurance-like Effects? *Strateg. Manag. J.* 2017, 38, 455–470. [CrossRef]
- Cheng, B.; Ioannou, I.; Serafeim, G. Corporate Social Responsibility and Access to Finance. *Strateg. Manag. J.* 2014, 35, 1–23. [CrossRef]
- Oh, W.Y.; Chang, Y.K.; Martynov, A. The Effect of Ownership Structure on Corporate Social Responsibility: Empirical Evidence from Korea. J. Bus. Ethics 2011, 104, 283–297. [CrossRef]
- 59. Udayasankar, K. Corporate Social Responsibility and Firm Size. J. Bus. Ethics 2008, 83, 167–175. [CrossRef]

60. Fang, M.; Nie, H.; Shen, X. Can Enterprise Digitization Improve ESG Performance? Econ. Model. 2023, 118, 106101. [CrossRef]

61. Sun, L.; Ashikin, N.; Saat, M. How Does Intelligent Manufacturing Affect the ESG Performance of Manufacturing Firms ? Evidence from China. *Sustainability* **2023**, *15*, 2898. [CrossRef]

- 62. Liao, L.; Lin, T.P.; Zhang, Y. Corporate Board and Corporate Social Responsibility Assurance: Evidence from China. *J. Bus. Ethics* **2018**, *150*, 211–225. [CrossRef]
- 63. Yuan, Y.; Tian, G.; Lu, L.Y.; Yu, Y. CEO Ability and Corporate Social Responsibility. J. Bus. Ethics 2019, 157, 391–411. [CrossRef]
- 64. Adams, R.B.; Ferreira, D. Women in the Boardroom and Their Impact on Governance and Performance. *J. Financ. Econ.* **2009**, *94*, 291–309. [CrossRef]
- Menicucci, E.; Paolucci, G. Board Diversity and ESG Performance: Evidence from the Italian Banking Sector. Sustainability 2022, 14, 13447. [CrossRef]
- Galbreath, J. When Do Board and Management Resources Complement Each Other? A Study of Effects on Corporate Social Responsibility. J. Bus. Ethics 2016, 136, 281–292. [CrossRef]
- 67. Tian, P.; Lin, B. Impact of Financing Constraints on Firm's Environmental Performance: Evidence from China with Survey Data. *J. Clean. Prod.* **2019**, 217, 432–439. [CrossRef]
- 68. Liang, H.; Renneboog, L. On the Foundations of Corporate Social Responsibility. J. Financ. 2017, 72, 853–910. [CrossRef]
- Hillman, A.J.; Keim, G.D. Shareholder Value, Stakeholder Management, and Social Issues: What's the Bottom Line? *Strateg. Manag. J.* 2001, 22, 125–139. [CrossRef]
- Crifo, P.; Diaye, M.A.; Pekovic, S. CSR Related Management Practices and Firm Performance: An Empirical Analysis of the Quantity-Quality Trade-off on French Data. Int. J. Prod. Econ. 2016, 171, 405–416. [CrossRef]
- 71. Tsai, H.J.; Wu, Y. Changes in Corporate Social Responsibility and Stock Performance. J. Bus. Ethics 2022, 178, 735–755. [CrossRef]
- 72. Benlemlih, M.; Bitar, M. Corporate Social Responsibility and Investment Efficiency. J. Bus. Ethics 2018, 148, 647–671. [CrossRef]
- Bouslah, K.; Hmaittane, A.; Kryzanowski, L.; M'Zali, B. CSR Structures: Evidence, Drivers, and Firm Value Implications. J. Bus. Ethics 2022. [CrossRef]
- Andersen, D.C. Do Credit Constraints Favor Dirty Production? Theory and Plant-Level Evidence. J. Environ. Econ. Manag. 2017, 84, 189–208. [CrossRef]
- Rennings, K. Redefining Innovation—Eco-Innovation Research and the Contribution from Ecological Economics. *Ecol. Econ.* 2000, *32*, 319–332. [CrossRef]
- 76. Acemoglu, D.; Akcigit, U.; Hanley, D.; Kerr, W. Transition to Clean Technology. J. Polit. Econ. 2016, 124, 52–104. [CrossRef]

77. Liu, J.Y.; Xia, Y.; Fan, Y.; Lin, S.M.; Wu, J. Assessment of a Green Credit Policy Aimed at Energy-Intensive Industries in China Based on a Financial CGE Model. *J. Clean. Prod.* **2017**, *163*, 293–302. [CrossRef]

- 78. Xu, X.; Li, J. Asymmetric Impacts of the Policy and Development of Green Credit on the Debt Financing Cost and Maturity of Different Types of Enterprises in China. *J. Clean. Prod.* **2020**, *264*, 121574. [CrossRef]
- 79. Deschênes, O.; Greenstone, M.; Shapiro, J.S. Defensive Investments and the Demand for Air Quality: Evidence from the NOx Budget Program. *Am. Econ. Rev.* 2017, 107, 2958–2989. [CrossRef]
- 80. Bjuggren, C.M. Employment Protection and Labor Productivity. J. Public Econ. 2018, 157, 138–157. [CrossRef]
- Tian, M.; Xu, G.; Zhang, L. Does Environmental Inspection Led by Central Government Undermine Chinese Heavy-Polluting Firms' Stock Value? The Buffer Role of Political Connection. J. Clean. Prod. 2019, 236, 117695. [CrossRef]
- Cui, J.; Zhang, J.; Zheng, Y. Carbon Pricing Induces Innovation: Evidence from China's Regional Carbon Market Pilots. AEA Pap. Proc. 2018, 108, 453–457. [CrossRef]
- Buchanan, B.; Cao, C.X.; Chen, C. Corporate Social Responsibility, Firm Value, and Influential Institutional Ownership. J. Corp. Financ. 2018, 52, 73–95. [CrossRef]
- Gilje, E.P.; Taillard, J.P. Does Hedging Affect Firm Value? Evidence from a Natural Experiment. *Rev. Financ. Stud.* 2017, 30, 4083–4132. [CrossRef]
- Wang, Z.; Liao, K.; Zhang, Y. Does ESG Screening Enhance or Destroy Stock Portfolio Value? Evidence from China. *Emerg. Mark. Financ. Trade* 2022, 58, 2927–2941. [CrossRef]
- Zhang, X.; Zhao, X.; He, Y. Does It Pay to Be Responsible? The Performance of ESG Investing in China. *Emerg. Mark. Financ. Trade* 2022, 58, 3048–3075. [CrossRef]
- Liu, P.; Zhu, B.; Yang, M.; Chu, X. ESG and Financial Performance: A Qualitative Comparative Analysis in China's New Energy Companies. J. Clean. Prod. 2022, 379, 134721. [CrossRef]
- Hadlock, C.J.; Pierce, J.R. New Evidence on Measuring Financial Constraints: Moving beyond the KZ Index. *Rev. Financ. Stud.* 2010, 23, 1909–1940. [CrossRef]
- Li, Q.; Wang, J.; Cao, G.; Zhang, J. Financial Constraints, Government Subsidies, and Corporate Innovation. *PLoS ONE* 2021, 16, e0259642. [CrossRef] [PubMed]
- 90. Yao, L.; Yang, X. Can Digital Finance Boost SME Innovation by Easing Financing Constraints?: Evidence from Chinese GEM-Listed Companies. *PLoS ONE* **2022**, *17*, e0264647. [CrossRef]
- Ju, X.S.; Lu, D.; Yu, Y. Financing Constraints, Working Capital Management and the Persistence of Firm Innovation. *Econ. Res. J.* 2013, 1, 4–16.
- 92. Zhang, S.; Wu, Z.; Wang, Y.; Hao, Y. Fostering Green Development with Green Finance: An Empirical Study on the Environmental Effect of Green Credit Policy in China. *J. Environ. Manag.* **2021**, *296*, 113159. [CrossRef]

- Pang, L.; Zhu, M.N.; Yu, H. Is Green Finance Really a Blessing for Green Technology and Carbon Efficiency? *Energy Econ.* 2022, 114, 106272. [CrossRef]
- Lu, Y.; Gao, Y.; Zhang, Y.; Wang, J. Can the Green Finance Policy Force the Green Transformation of High-Polluting Enterprises? A Quasi-Natural Experiment Based on "Green Credit Guidelines". *Energy Econ.* 2022, 114, 106265. [CrossRef]
- 95. Wu, J. Financing Constraints for R&D in China: The Role of State Ownership. Appl. Econ. Lett. 2019, 26, 1162–1166. [CrossRef]
- 96. Jin, Y.; Gao, X.; Wang, M. The Financing Efficiency of Listed Energy Conservation and Environmental Protection Firms: Evidence and Implications for Green Finance in China. *Energy Policy* **2021**, *153*, 112254. [CrossRef]
- 97. Zhang, D.; Du, W.; Zhuge, L.; Tong, Z.; Freeman, R.B. Do Financial Constraints Curb Firms' Efforts to Control Pollution? Evidence from Chinese Manufacturing Firms. J. Clean. Prod. 2019, 215, 1052–1058. [CrossRef]
- Luo, X.; Wang, H.; Raithel, S.; Zheng, Q. Corporate Social Performance, Analyst Stock Recommendations, and Firm Future Returns. *Strateg. Manag. J.* 2015, 36, 123–136. [CrossRef]
- Marquis, C.; Qian, C. Corporate Social Responsibility Reporting in China: Symbol or Substance? Organ. Sci. 2014, 25, 127–148. [CrossRef]
- Wang, Z.; Reimsbach, D.; Braam, G. Political Embeddedness and the Diffusion of Corporate Social Responsibility Practices in China: A Trade-off between Financial and CSR Performance? J. Clean. Prod. 2018, 198, 1185–1197. [CrossRef]
- 101. Lau, C.M.; Lu, Y.; Liang, Q. Corporate Social Responsibility in China: A Corporate Governance Approach. J. Bus. Ethics 2016, 136, 73–87. [CrossRef]
- McGuinness, P.B.; Vieito, J.P.; Wang, M. The Role of Board Gender and Foreign Ownership in the CSR Performance of Chinese Listed Firms. J. Corp. Financ. 2017, 42, 75–99. [CrossRef]
- Li, P.; Lu, Y.; Wang, J. Does Flattening Government Improve Economic Performance? Evidence from China. J. Dev. Econ. 2016, 123, 18–37. [CrossRef]
- Fafaliou, I.; Giaka, M.; Konstantios, D.; Polemis, M. Firms' ESG Reputational Risk and Market Longevity: A Firm-Level Analysis for the United States. J. Bus. Res. 2022, 149, 161–177. [CrossRef]
- 105. Invesco. China to See Significant Growth in Responsible and ESG-Themed Investment. 2021. Available online: https://www.invesco. com/apac/en/institutional/insights/esg/China\_to\_see\_significant\_growth\_in\_responsible\_and\_ESG-themed\_investment.html (accessed on 22 March 2021).
- 106. PwC. ESG-Focused Institutional Investment Seen Soaring 84% to US\$33.9 Trillion in 2026, Making up 21.5% of Assets under Management: PwC Report. 2022. Available online: https://www.pwccn.com/en/press-room/press-releases/pr-131022.html#:~: text=Beijing%2C%2013%20October%202022%20-%20Asset%20managers%20globally,total%20global%20AuM%20in%20less% 20than%205%20years (accessed on 13 October 2022).
- 107. The Asset. Sustainable Investing Enters the Fast Lane in China. 2022. Available online: https://www.theasset.com/article-esg/ 45838/theasset.com (accessed on 12 January 2022).
- 108. Pástor, Ľ.; Stambaugh, R.F.; Taylor, L.A. Sustainable Investing in Equilibrium. J. Financ. Econ. 2021, 142, 550–571. [CrossRef]

**Disclaimer/Publisher's Note:** The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.